Title: Gestational weight gain information: Seeking and sources among pregnant women

Authors:

Jane C Willcox (jwillcox@deakin.edu.au)
Karen J Campbell (kcamp@deakin.edu.au)
Elizabeth A McCarthy (eamcca@unimelb.edu.au)
Martha Lappas (mlappas@unimelb.edu.au)
Kylie Ball (kylie.ball@deakin.edu.au)
David Crawford (david.crawford@deakin.edu.au)
Alexis Shub (ashub@internode.on.net)
Shelley A Wilkinson (Shelley.Wilkinson@mater.org.au)

Version: 5 Date: 1 July 2015

Author's response to reviews:

Letter attached